Author's response to reviews

Title: Use of drug therapy in the management of symptomatic ureteric stones in hospitalised adults: a multicentre placebo controlled randomised trial of a calcium channel blocker (nifedipine) and an alpha-blocker (tamsulosin): study protocol of the randomised controlled SUSPEND trial

Authors:

Sam McClinton (s.mcclinton@nhs.net)
Kathryn Starr (k.starr@abdn.ac.uk)
Ruth Thomas (r.e.thomas@abdn.ac.uk)
Graeme McLennan (g.maclennan@abdn.ac.uk)
Gladys McPherson (g.mcpherson@abdn.ac.uk)
Alison McDonald (a.mcdonald@abdn.ac.uk)
Thomas Lam (thomasblam@abdn.ac.uk)
James N'Dow (j.ndow@abdn.ac.uk)
Mary Kilonzo (m.kilonzo@abdn.ac.uk)
Robert Pickard (r.s.pickard@ncl.ac.uk)
Ken Anson (ken.anson@stgeorges.nhs.uk)
Jennifer Burr (jmb28@st-andrews.ac.uk)

Version: 2 Date: 31 October 2013

Author's response to reviews:

We have added details of ethic committee approval and stated that the trial complies with the Declaration of Helsinki.

Additional documents have been removed as requested.